Project acronym: Project full title: WASP (Wind Assisted Ship Propulsion) Run Wind Propulsion Technology real-life trials on sea-going ships in operation>showcase proven concepts>market adoption>green sea transport 38-2-6-19 Netherlands Maritime Technology Foundation

Project No. Coordinator:









D 5.A.1

Scenario development for wind propulsion technology adoption: A theoretical model for agent-based modeling

Deliverable data

Dellesseehte Ne			
Deliverable No	5.A.1		
	Scenario development for wind propulsion te	chnology adoption: A	
Deliverable Title	theoretical model for agent-based modeling		
Work Package no: title	WP 4 Policy and viable business:		
Deliverable type	Report/strategy		
Lead beneficiary	NUY		
Responsible author	Roberto Rivas Hermann		
	Ning Lin		
Co-authors	Mark David Siatta		
Date of delivery	31-03-2021		
Approved	Name	Date	
Peer reviewer 1	Vasileios Kosmas (KLU)	16-04-2021	
Peer reviewer 2	Lara Pomaska (KLU)	16-04-2021	
Peer reviewer 3	Kasper Uithof (NMTF)	19-04-2021	

Document history

Version	Date	Description
1	08-04-2021	First draft
2	25-05-2021	Amendments following review
3	27-05-2021	Final version



Contents

1.	Intr	oducti	on	8
	1.1	Purpo	ose of the report, contributions and structure	9
2.	Me	thodol	ogy	11
	2.1	Phase	e 1: Technical requirements of the ABM methodology	12
	2.2 mariti	Phase me inc	e 2: Literature review on the drivers of eco-innovation adoption, with a focus o lustry	on the 12
	2.3	Phase	e 3: Case study of the drivers of WPT adoption by short sea shippers	14
	2.3	.1 S	Survey with WPT stakeholders	14
	2.3	2 F	ocus group with shipping companies installing WPTs	15
	2.4	Phase	e 4: Model development	15
3.	An	ABM fr	amework to model WPT adoption in short sea shipping	16
	3.1	Devel	oping a theoretical model in ABM	16
	3.2	ABM	applications in the shipping industry and eco-innovations	17
4.	Driv	vers of	eco-innovation and the micro-economics of shipping firms	19
	4.1	Drive	rs for the adoption of WPT: Four theoretical propositions	19
	4.2	Econo	omics of shipping firms' operations	22
5.	Em	pirical	insights into WPT adoption drivers in the shipping industry	23
	5.1	Busin	ess case workshop	23
	5.2	Surve	y results	25
	5.2	.1 E	nvironmental performance	25
	5.2	.2 F	Regulatory push	26
	5.2	.3 N	/larket pull	26
	5.2	.4 T	echnology push	27
6.	Pro	posed	theoretical model of WPT adoption in shipping	29
	6.1	Choo	sing the research question	29
	6.2	Choo	sing mobile agents	29
	6.3	Choo	sing mobile agent properties	29
	6.4	Choo	sing environmental agents	30
	6.5	Choo	sing agent behaviors	30
	6.6	Desig	ning a time step	32
	6.7	Defin	ing the parameters of the model	32
	6.8	Defin	ing measures	32
7.	Cor	nclusio	n	34



Appendix 1 – Summary of reference ABM on transportation eco-innovation. Source: Zhang et a	al.
Appendix 2- Summary of reference ABM on WPT diffusion. Source: Karslen et al. (2019)	59 42
Appendix 3- Survey with experts	46

Figures

Figure 1 Systemic combining approach to developing a theoretical model of WPT	adoption in
shipping firms	11
Figure 2 The four phases of the systemic combination process	12
Figure 3 Overview of the systematic literature review process on the drivers of ec	o-innovation
adoption. Adapted from Bossle et al. (2016)	13
Figure 4 Process of ABM. Source: (Garcia, 2007)	16
Figure 5 Time steps of the interval	32

Tables

Table 1 Relationship between the guiding research question, context, and case study......14



Acronyms and Abbreviations

- ABM Agent-based model
- CO₂ Carbon dioxide
- EU European Union
- GHG Greenhouse gas(es)
- IMO International Maritime Organization
- LNG Liquified natural gas
- NOx Nitrogen oxide
- SME Small and medium enterprises
- WASP Wind assisted ship propulsion Interreg North Sea Region Project
- WPT Wind propulsion technology



Executive Summary

Purpose of this report

This report is the first deliverable of activity 5a, WP4 – Policy and Viable Business of the WASP Project. The objective of the report is to provide a theoretical basis for a model of wind-assisted ship propulsion technology (WPT) adoption by shipowners.

The answer to the question of why shipowners would choose to install WPT will serve as a basis for the development of an agent-based model (ABM) of WPT adoption (activity 5a, report 2). There is thus a close relationship between both reports. This report will thus focus on the model and its theoretical foundations, while report 2 will focus on the model's empirical basis and use in simulations. For the preparation of report 2, Nord University will collaborate with an external consultant, primarily to support researchers with the programming aspects linked to the development of the ABM.

Methodological approach

This report has been prepared using a combined systemic/inquiry approach. This approach suits the topic as it supports an interactive cycle of theoretical development incorporating the specific case of WPT adoption, eco-innovation adoption in shipping more broadly, ABM methodologies in eco-innovation and shipping, and the empirical context of the WASP Project. The systemic/inquiry strategy allows a theoretical model to be refined as the basis for the ABM.

The systemic/inquiry approach we have employed is comprised of the following elements:

i) Theory; literature review of the drivers of eco-innovation.

WPTs are part of a larger category of eco-innovations. This report thus incorporates a literature review on the drivers of the adoption of eco-innovations to understand the factors prompting maritime firms to adopt green innovations (including both products and processes). We also review existing literature about maritime business, with particular emphasis on shipping firms' cost structures and the decision drivers behind operational and technological upgrades.

ii) Framework; ABM methodology as an approach to the adoption of WPTs.

We also introduce a methodology for developing agent-based models, with a particular focus on theoretical model development. Existing ABMs which focus on transportation eco-innovations and WPT are reviewed.

iii) The case; understanding the shipping industry's expectations towards WPTs

Based on the literature review, we identify drivers prompting shipping companies to adopt WPTs, and report the results of a survey conducted with practitioners and experts to rate these drivers and determine the key factors to include in the theoretical model.

In parallel with this survey, during spring 2020, the WASP Project organized a business case workshop, which provided valuable inputs from the shipowners involved in the WASP Project.



Findings

With the growth of international trade volumes, the environmental impact of the shipping industry has increased significantly. The International Maritime Organization is expected to adopt stricter policies and encourage the shipping industry to become more environmentally friendly. Wind-assisted ship propulsion technology has the potential to reduce ship emissions dramatically. This paper aims to model shipowners' installation of different types of WPT over time, and the relationships between adoption of WPT and market segments, geographical scope, oscillating markets, and regulatory conditions. By conducting this agent-based modeling research, we aim to identify the diffusion speed of different WPTs under a range of business and policy conditions.

An agent-based model is developed using the methodology discussed in Section II. Based on the research question, the authors propose a model to simulate the decision-making process of shipowners, regarding the installation and use of WPT on their ships. There are four mobile agents in the model, including shipowners and the technology providers for Flettner rotors, Ventifoil, and fixed wing rigs. The model includes two general properties for both shipowners and technology providers, namely location and actor-direction in the model ('heading'), and two more properties for shipowners, namely available cash and monthly growth of capital. We also include two environmental agents: market situation and regulatory condition. These are defined, respectively, in terms of the change rates of fuel prices and fuel switching costs.

In this shipowner-technology provider model, we define a timeline of behaviors for each agent which can help readers and programmers to understand how agents interact with each other and with business environments. This timeline is as follows: (1) shipowners and technology providers move throughout the simulating software's space; (2) shipowners interact with shipowners and technology providers; (3) update and check confidence level of shipowners on the new technology; (4) shipowners check cash availability; (5) shipowners decide whether or not to install WPT devices; (6) shipowners install WPT devices; (7) shipowners update confidence level and cash availability; and (10) fuel prices change.

Two parameters are included in the model: (1) initial cash available and (2) initial confidence level. These two parameters enable us to see how different initial situations influence the diffusion of WPT. Considering the purpose of this study, we use the following measures for each studied ship type: (1) the percentage of ships with fixed wing rigs, Flettner rotors, and Ventifoils installed over time; (2) the percentage of ships with these technologies in active use, over time; and (3) oil prices over time. These measures are examined under a variety of scenarios for different ship types in order to illustrate WPT diffusion over time.



1. Introduction

The maritime industry is seeking solutions to mitigate greenhouse gas (GHG) emissions, air pollution, and other climate impacts (Hermann, 2017). At present, numerous options such as energy efficiency improvements, renewable energy sources (including wind propulsion), liquified natural gas (LNG), and emissions reduction technologies (e.g., scrubbers, carbon capture and storage) are being considered and implemented to transition the industry towards a low-carbon future (Rehmatulla, Calleya, et al., 2017).

Several recent and ongoing technological developments in the shipping industry align well with the concept of eco-innovation, which refers to the process of developing and introducing new products, processes and services that, once introduced, lead to positive change in an organization's environmental impact (Rennings, 2000).

Wind energy has been a key source of propulsive power for millennia, although in recent times, despite its status as an abundant and free renewable energy resource, it has not been adequately utilized in the shipping industry (Talluri et al., 2016). Wind propulsion technologies (WPT) are promising options among renewable energy sources for maritime use (Karslen et al., 2019), and are designed to reduce fuel consumption, thus decreasing costs while also cutting emissions. The first research output of WP4 in the WASP Project provides a good overview of the recent technological development of WPT (Chou et al., 2021)

The extant literature on eco-innovation includes a significant stream of research on the determinants of eco-innovation. This research has primarily been concerned with the interaction of public policies, market incentives, and characteristics of businesses in driving the adoption of eco-innovations (F. Boons & McMeekin, 2019). As a result, the literature has provided illustrative cases of the adoption of eco-innovation in different industry sectors, including the maritime (Rivas-Hermann et al., 2015), automotive (Christensen, 2011), agriculture (Hasler et al., 2016), and construction (Zabalza Bribián et al., 2011) industries.

In general, these studies tend to connect eco-innovation processes with the interactions between driving factors such as market pull, regulatory push/pull, and technological push (Kesidou & Demirel, 2012). Market pull is generally understood as the impact of consumer choices that incentivize research into and the development of greener technologies. Regulatory push/pull describes the standards, policies, regulations, and laws administered to address environmental impact. Technological push refers to industry-specific processes, practices, and operations (Horbach et al., 2012; Kesidou & Demirel, 2012; Rennings, 2000). Recent findings have shown that these drivers catalyze operations to exploit opportunities stemming from new regulations, market leanings, and technologies (Díaz-García et al., 2015). Meanwhile, multi-sectoral studies within countries have analyzed what drives companies to adopt eco-innovations through quantitative methods and large samples (Bossle, De Barcellos, et al., 2016; Cuerva et al., 2014; Kesidou & Demirel, 2012).

Overall, the eco-innovation studies highlighted above indicate that sectoral characteristics play an important role in the diffusion and adoption of eco-innovations. The normative characteristics of a given sector thus play a role in understanding the context in which firms opt for the adoption of eco-innovations (Horbach, 2019). While a wide variety of methodological approaches have been used to understand the micro-economics of a firm's eco-innovation adoption (Bossle, Dutra de Barcellos, et al., 2016), modeling and simulation methods appear particularly suited to capturing the complex and changing environments in which firms decide to adopt eco-innovations (Köhler, 2019).



Among these modeling and simulation methodologies, agent-based modeling (ABM) is quite popular among researchers. ABM is a computational methodology with origins in the modeling of complex systems. A key characteristic is the presence of agents, computational entities with defined properties ,such as predicted actions in a given set of circumstances, e.g. the decision on whether to invest. The behavior of each agent can be modified by adjusting programmed rules; therefore, multiple scenarios can be modelled based on predefined instructions that respond to theoretical models (Wilensky & Rand, 2015).

Zhang et al. (2011) relied on ABM to understand the impact of public policies on consumer choices for the adoption of the eco-innovation of electrical cars. In an ABM study of socio-technological change in shipping, Karslen et al. (2019) modeled the influence of imperfect agent information and split incentives using a case study of Flettner rotors as a form of WPT. The focus of this study was, however, focused on climate-energy policies.

As these studies show, although ABMs are not the most frequently used methodology in ecoinnovation research, they are often used to answer questions related to the interaction of multiple drivers and the role of these drivers in focal firms' (i.e., agents') decisions regarding adoption of new technologies. ABMs' ability to simulate whole systems with multiple interacting agents can provide potential insight into the external and firm-specific conditions under which diffusion of WPTs is likely.

1.1 Purpose of the report, contributions and structure

This report, as an output of the WASP (Wind Assisted Ship Propulsion) Project, addresses the following research questions:

- 1. Can an ABM-focused theoretical framework be applied to develop a model of ecoinnovation (WPT) adoption in the maritime shipping industry?
- 2. What drivers motivate shipping firms to adopt eco-innovations in general, and how do micro-economic conditions influence these decisions?
- 3. How do external and internal driving factors influence the adoption of WPTs by shipping firms in the North Sea Region?
- 4. How do these different factors interact with each other in an ABM?

This report is the first of two deliverables in *WP4 Policy and Viable Business, activity 5a: Scenario development for WPT market uptake.* A more detailed overview of WPTs is available in the WASP article (Chou et al., 2021). The first deliverable develops a theoretical model for ABM simulation, which is subsequently implemented, calibrated, and validated in report #2.

The remainder of the report is structured in the following sections:

- Section II presents the methodology along with a discussion of the empirical data collected.
- Section III addresses the question: "Can an ABM-focused theoretical framework be applied to develop a model of eco-innovation (WPT) adoption in the maritime shipping industry?". This section reviews the key tenets of the ABM methodology as well as the previous implementation of ABM in both eco-innovation and the shipping industry.
- Section IV summarizes the analytical framework, and provides a detailed review of the drivers for eco-innovation and shipping firms' operational economics (research question 2). This section proposes a set of likely drivers for eco-innovation in the adoption of WPTs in shipping.
- Section V presents an empirical look into the adoption drivers of WPTs in the shipping industry (research question 3).



- Section VI combines the main inputs of Sections II to V to propose an ABM-based theoretical framework for WPT adoption by shipping firms (research question 4)
- Section VII presents conclusions and frames the terms for the follow-up report as part of WP4 of the WASP Project.



2. Methodology

In this report, we seek to explain the potential drivers influencing a firm's decision to adopt WPT, and the connections between internal and external drivers. The research presented here is explanatory in nature, as it aims to i) explain patterns related to the phenomenon in question and ii) identify relationships shaping the phenomenon (Marshall & Rossman, 2006). We rely on abductive inference in formulating new ideas and analytical approaches without drawing on preliminary theoretical premises (Meyer & Lunnay, 2013). Abductive inference is characterized by generating alternative explanations for data that does not fit the expected theoretical propositions (Meyer & Lunnay, 2013). We adapt the abductive methodology known as systemic combining (Dubois & Gadde, 2002), which repeatedly adjusts theoretical lenses and methodological approaches through an interactive approach to better respond to the characteristics of the empirical world. Systemic combining is also used when carrying out complex case studies which require the researcher to have a deep understanding of organizational dynamics. We can thus strategically use systemic combining as a way to connect theoretical framing with empirical data and a specific case study (Dubois & Gadde, 2002). Figure 1 visualizes the systemic combining process. This study was carried out in four phases, as further described in Figure 2.



Figure 1 Systemic combining approach to developing a theoretical model of WPT adoption in shipping firms





Figure 2 The four phases of the systemic combination process

2.1 Phase 1: Technical requirements of the ABM methodology

In Phase 1, we conduct a review of the technical requirements in designing an ABM. ABMs are programmed using computational rules; in this case, these rules are based on the factors we identify as drivers in agent decisions (Wilensky & Rand, 2015). Section III of this report reviews previous ABMs in both eco-innovation and shipping research to understand data needs, agent definitions, ABM process flows, and overall software requirements. The purpose of Phase 1 is not to provide a lengthy or technically intensive presentation of ABM, but to highlight how theoretical models for running ABM should be developed.

2.2 Phase 2: Literature review on the drivers of eco-innovation adoption, with a focus on the maritime industry

Phase 2 consists of a systematic literature review with a focus on the drivers of adoption of ecoinnovations. The purpose of such a review is to provide a reliable and reproducible synthesis of key thematic discussions within a given academic field (Easterby-Smith et al., 2018). A systematic review in the context of eco-innovation is primarily concerned with identifying primary drivers, motivations, and barriers that have been shown to lead firms to either adopt or not adopt ecoinnovations. Following Bossle et al. (2016), the literature review on eco-innovation drivers was organized according to the elements highlighted in Figure 3.





Figure 3 Overview of the systematic literature review process on the drivers of eco-innovation adoption. Adapted from Bossle et al. (2016)

The starting point in the systematic literature review process is the guiding research question, which is formulated as follows: *What mix of eco-innovation drivers higher adoption rates of WPT in the shipping industry?*

Based on the guiding research question, articles were selected for inclusion in the literature review based on a seven-step process. In step 1, keywords were identified by consulting previous systematic literature reviews (Díaz-García et al., 2015). Articles relevant to the topic were then located in the Scopus database using these keywords; Scopus was chosen due to its repository of relevant high-quality research. We carried out a preliminary inspection of previous reviews on "innovation" and "eco-innovation". Subsequently, we broadened the search to also include other search terms, such as "adoption" OR "drivers", and to include other qualifiers such as "innov*" and "eco-innov*". The results from this search were then used to create a final list of key terms to cover all possible existing literature relating to eco-innovation and the underlying drivers that catalyze adoption rates.

A set of exclusion criteria was established in step 2, including temporal criteria, as the focus was on the most recent academic contributions (2013-2020). In step 3, we identified research focused on the discussion of eco-innovation within three key journals of technology and environmental management: Journal of Cleaner Production, Ecological Economics, and Journal of Engineering and Technology Management. In step 4, a detailed analysis of the titles and abstracts of candidate articles in connection with the research question resulted in a preliminary list of 13 articles for further inclusion.

The selected academic literature was complemented with industry-driven articles derived from the shipping magazine Lloyd's List. These articles were included to analyze the industry consensus on eco-innovations and how it is being discussed by industry insiders. The query used keywords such as "clean technology", "eco-technology", "green innovation" and "wind propulsion technologies" to identify the articles to be used. This search returned 47 mentions of the above terms in various articles dating back to the year 2008. These articles were utilized as an industry-centric counterbalance to the academia-focused systematic literature review on eco-innovation.



Academic literature and industry articles were analyzed critically to prepare a summary leading to the set of propositions summarized at the end of Section IV. These propositions were subsequently integrated into the survey in Phase 3.

This systematic review was complemented with a review of the literature on micro-economic decisions in shipping firms. The combination of the two reviews provides an initial theoretical understanding of the factors that influence the likelihood that shipping companies will adopt WPTs.

2.3 Phase 3: Case study of the drivers of WPT adoption by short sea shippers

Stage 3 follows a single case study design. The purpose of this case study is to tackle research question 3: "How do external and internal driving factors influence the adoption of WPTs by shipping firms in the North Sea Region?"

Phase 2 developed the theoretical framework which is necessary to analyze the single case (Yin, 2018). The theoretical framework also provides inputs for the data collection instruments used in Phase 3, which are summarized in Table 1 below. The case study focuses on WPT adoption drivers in the short sea shipping sector, mainly within the North Sea Region.

Guiding issue	Method of data collection	Section in the report
Internal drivers of WPT adoption: Why do shipping companies install WPTs?	Focus group with shipping companies installing WPTs; business case workshop	Section V
Validation of the key eco- innovation drivers as applied to WPTs	Survey with WPT stakeholders	Section V

 Table 1 Relationship between the guiding research question, context, and case study

2.3.1 Survey with WPT stakeholders

The survey was designed to gather information on the driving factors of eco-innovative behavior and performance, and comprised ten questions relating to the various identified drivers, including technology push, regulatory pull, and market push. The questions adapted the theoretical propositions resulting from Phase 3 of the study (Appendix 3). The participants were asked to respond to all ten questions using a five-point Likert scale ranging from 1 = Strongly Disagree to 3 = Do Not Agree or Disagree to 5 = Strongly Agree.

The survey respondents were eco-innovation practitioners with business, education, industrial, or research backgrounds. They were identified and recruited in two ways. One group of participants had previously been involved with the WASP Project and or EU-backed shipping demonstration initiatives; these individuals were deemed suitable due to their knowledge of the shipping industry in general, prior experience with shipping eco-innovations, and commitment to fostering sustainable development as signified by their involvement in previous demonstration projects. The other group was recruited from among the authors of academic articles pertaining to innovation, eco-innovation, green innovation, WPT, and the shipping industry. It was deemed necessary to expand the scope of participants beyond past participants in related projects to ensure a wide population of competent and knowledgeable individuals.

Participants represented a number of countries, including Belgium, Brazil, Denmark, France, Germany, Italy, the Netherlands, Norway, Portugal, Spain, Sweden, Turkey, and the United States. These participants held positions in educational institutions, governmental organizations, shipping



companies, and consultancies; their job titles included professor, researcher, head of research unit, port logistics director, dean, special projects manager, director, consultant, and CEO. From a sample of 111 selected participants, 11 could not be contacted due to old, expired, or malfunctioning e-mail addresses. Of the remaining 100, 19 respondents completed the survey, representing a response rate of 19%.

2.3.2 Focus group with shipping companies installing WPTs

In collaboration with WASP Project partners, we held a focus group on 11.05.2020. This focus group was conducted virtually due to the COVID-19 restrictions. It included 16 participants, including two shipping companies participating in the WASP Project. The focus group was organized into three parts:

- 1. The business problem/opportunity and the operational environment
- 2. Benefits of WPT technologies versus challenges/risks
- 3. The cost factors

2.4 Phase 4: Model development

Phase 4 involved the development of the ABM theoretical model (Figure 2). It was directly linked to the inputs provided by Phases 1, 2 and 3. Phase 1 identified the technical requirements, including an overview of previous ABM models of eco-innovation in transportation and the type of data requirements. Phases 2 and 3 will, on the other hand, adapt the specific conditions of WPT adoption by shipping firms and align these with the requirements established in Phase 1.

We have developed the ABM model for this project through an exploratory process involving the definition of agents and behaviors and subsequent exploration of emerging patterns (Wilensky & Rand, 2015).



3. An ABM framework to model WPT adoption in short sea shipping

This section describes the agent-based model and the logic of the associated theoretical framework. The use of agent-based modeling is gaining popularity across a range of academic subjects in both the natural and social sciences (Wilensky & Rand, 2015). In the social sciences, ABM is used to model the behavior of individual or group actors in dynamic adaptive systems; these systems can take the form of marketplaces, organizations, or other systems that can be analyzed via the combined actions of large groups of individuals (Garcia, 2005).

In an ABM approach, agents are autonomous decision-making entities that interact with each other and/or their environment based on specific rule sets which dictate their behavior and choices (Zhang et al., 2011). This methodology is a useful tool to understand eco-innovation and new product development, specifically in the shipping industry, because the process of innovation, imitation, and enhancement is not deterministic, and thus the system being modeled is highly non-linear (Debenham & Wilkinson, 2006). ABM allows for detailed examination of the full range of actor-specific strategies during the innovation process.

Utilizing external knowledge sources is a vital product development strategy, which is why innovation benefits from collaborative partnerships within innovation networks through the dynamic and heterogeneous unit nexuses, because each part harnesses different collections of expertise (Gilbert et al., 2007).

The development of ABM occurs in three broad phases (Garcia, 2005, 2007; Wilensky & Rand, 2015): theoretical model development, simulation in ABM software, and validation with empirical data. The last step in this sequence ensures quality of results and allows the empirical calibration of the model (Figure 4).



Figure 4 Process of ABM. Source: (Garcia, 2007).

3.1 Developing a theoretical model in ABM

Garcia (2005)'s widely cited paper on the application of ABM in innovation management models exploration and exploitation mechanisms at the firm level as a way to help an organization's managers make decisions about investments in new product development portfolios. The conceptual model includes six steps in the development of ABMs, starting with the definition of a



research question that will subsequently guide the ABM process. These steps and key considerations of each are listed below:

Step 1. Theory operationalization

- Involves creation of a cognitive map
- Simple; the goal is to determine the allocation of resources for exploration or exploitation

Step 2. Agent specification

• The behavioral state of each individual: characteristics of the degree of relation to other agents (early adopter versus late adopter)

Step 3. Environmental specification

- Environments may be modeled as closed or open; open environments are those altered by events outside of the modeled system
- Implications for model development in the WASP Project context include fuel price/market scenarios

Step 4. Rules of behavior

- Both agent-agent rules and agent-environment rules must be established
- Implications for model development: What conditions will influence a shipowner to install/not to install a WPT? What is the role of charterers, government, and WPT manufacturers?

Step 5. Decisions regarding measurement of results

• Decisions include which type(s) of data to record, on what scale to measure this data, and what format results will be displayed or recorded in (e.g., figures, maps, or tables).

Step 6. Runtime specifications

• Decisions must be made regarding the number of iterations to run in order to properly model the system.

Specifications are based on interview feedback and the analysis of other qualitative data sources. The theoretical model development process synthetized above reflects that of other more generalist methodologies, such as the approach proposed by Wilensky and Rand (2015).

3.2 ABM applications in the shipping industry and eco-innovations

Several publications focusing on the application of ABM within the maritime transportation field were identified. A number of these publications model the impact of policy instruments, such as the sulfur cap on maritime fuels (Holmgren et al., 2014; Vierth et al., 2015). Holmgren et al. (2014) relied on an ABM developed as part of the EU-funded SKEMA project and applied the TAPAS tool to model decision-making processes in transportation chains, with a focus on high-value goods.

We identified two studies focusing on research questions similar to those of this report, i.e. drivers for the adoption of eco-innovations or cleaner technologies in the transportation sector (Karslen et al., 2019; Zhang et al., 2011). The details on both models are presented in Appendix 1 and 2, respectively.

Zhang et al.'s (2011) model focuses on the diffusion of electric vehicles (EV) as an eco-innovation. The aim of this ABM study is: "We focus the study on the impact of three specific mechanisms on



the diffusion of AFVs: technology change, consumer interactions, and regulatory policies" (Zhang et al, 2011, p. 153). " $\!\!\!$

The theoretical framing encompasses several theories connected to the drivers of eco-innovation, including technological push, regulatory push, and market pull (see the next section for a more detailed presentation of how these drivers influence WPT adoption). In addition, the ABM accounts for the characteristics of the different agents involved in the development and diffusion of EVs, including car manufacturers. The parameters in the ABM evaluate different EV designs, their acceptance by consumers, and how different policy combinations affect EV designs. The authors use this model to run three different experiments drawing on the eco-innovation model and the agent parameters.

Another ABM model developed by Karlsen et al. (2019) focuses on the diffusion of a specific type of WPT called Flettner rotors. The model answers questions concerning the effects of shipping firm participation in learning networks and pilot projects relating to the installation of WPTs, along with the effects of imperfect information or split incentive barriers and technology providers' interaction with shipowners. Key properties of the environment include fuel prices. The model incorporates several simplifications of the inherent characteristics of ABM to avoid an overwhelming model complexity: focus is narrowed to a single WPT technology and supplier, a limited number of policies (e.g. CO₂ pricing); impacts on shipping firms' adaption of WPT are examined only for one type of ship; and the cost calculation is limited to the case of a single long-distance deep-sea shipping route (Rotterdam to Shanghai). The model is also constrained to 100 shipowners and 100 model runs. In addition, to contribute to the simplification, equipment failures and catastrophes are assumed to be non-existent in the model.



4. Drivers of eco-innovation and the micro-economics of shipping firms

This section of the report summarizes Phase 2: Literature review of the drivers of eco-innovation and the micro-economics of shipping firms.

4.1 Drivers for the adoption of WPT: Four theoretical propositions

Proposition 1: Due to the dynamics of short sea shipping, regulatory measures will be most effective driver for the adoption of WPT

Regulations have historically been emphasized as an important driver of eco-innovation due to their two-fold effects on the environment and technical upgrades (Rennings, 2000). Particularly in the shipping industry, it has been observed that a mutual reinforcement relationship between *economic market pull* and *self-regulatory instruments* stimulating eco-innovation. However, due to the challenges of current market circumstances, shipowners are not sufficiently incentivized to install less environmentally damaging technologies (Hermann, 2017). It has also been shown that *technological push* and *regulatory push* are closely linked as eco-innovation drivers due to technological suppliers' lobbying of regulators to ensure certain technologies continue to be included in regulatory measures (Hermann, 2017).

Previous research on WPTs and regulations indicate that if a CO₂ taxation policy were to be implemented, the use of Flettner rotors in conjunction with diesel propulsion would provide an economic benefit. Conversely, in a scenario with no CO₂ taxation policy in place, the use of Flettner rotors on ships would provide little to no economic benefit for certain routes (Talluri et al., 2018). Another study (Talluri et al., 2016) concluded that the installation of vertical axis wind turbines or Flettner rotors could provide significant benefits on ships crossing the Atlantic Ocean because these routes are characterized by high fuel consumption and, more importantly, particularly strong wind conditions.

Karslen et al. (2019) suggest that improvements in the eco-innovation process require more stringent regulatory frameworks, including new standards and limits on access to existing subsidies and fiscal incentives; options include carbon taxation, which would directly impact the shipping sector. Additionally, after a period of carbon taxation and demonstration programs, a cost-effective policy could enhance the diffusion rates of rotor technology and reduce the need for this carbon taxation. A policy focus on routes with favorable wind conditions (e.g., Atlantic routes), forming naturally protected innovation niches would lead to learning, reduced barriers, and lower costs, thereby offering more benefits for shipowners operating on routes with less favorable wind conditions (Karslen et al., 2019).

Proposition 2: To stay competitive, once the first firm adopts, others will follow

There have been several studies showing a positive correlation between eco-innovation and a firm's economic performance (Cai & Li, 2018; Hojnik & Ruzzier, 2016). Firms that apply ecoinnovation in their business models generate greater revenue per employee than those that do not (Doran & Ryan, 2012). These benefits are seen primarily in operational economies, through cost savings and higher levels of resource productivity along with more efficient logistics and the subsequent commercialization (Sarkar, 2013). Furthermore, Cheng and Shiu (2012) found positive correlations between a firm's performance and its focus on eco-innovations in products, processes, and organization. Chen et al. (2017) and Cruz-Cázares et al. (2013) showed how innovation focusing



on technological efficiency is also positively linked to firm performance. Other studies have demonstrated that environmental innovation (i.e. the optimization of a firm's use of natural resources in terms of energy consumption per output unit) yields significant improvement on competitiveness and business performance (Bossle, Dutra de Barcellos, et al., 2016; Eiadat et al., 2008; Tseng et al., 2013).

Additionally, investments in eco-innovation have been shown to spawn an iterative cycle wherein firms' investment in eco-innovation leads to greater competitive advantages on the market (Hojnik & Ruzzier, 2016). This also supports the notion that firms that seek opportunities while maintaining open collaboration with other players in the market tend to develop eco-innovations at a higher rate. Firms that excel at recognizing opportunities engage in greater eco-innovation, and these firms also tend to have customer and market orientations that enable them to accurately identify consumer preferences and evolve their business models to further reinforce eco-innovation development (Sáez-Martínez et al., 2016). Other market-based benefits of the development and implementation of eco-innovation include improvement in the firm's image, cost savings, local community outreach improvement, broadened access to green markets, and further competitive advantages (Shrivastava, 1995). Furthermore, Sáez-Martínez (2016) indicated that path dependency is highly prevalent in eco-innovation development: firms with high levels of innovation development capacity are more likely to experience continued eco-innovation in the future. Thus, within the shipping industry, it can be surmised that WPTs have the potential to provide this kind of benefit, and with their expected cost savings of up to 20% (Talluri et al., 2018), this makes them extremely attractive. Even before this analysis, as far back as 2013, Scandlines implemented a longterm vision to incorporate hydrogen, Flettner rotors, and other environmental improvements into concept vessels to lure both potential buyers and customers who are seeking ways to cut their fuel consumption (Eason, n.d.).

Proposition 3: Technological push will be the least important factor because shipowners purchase WPT and do not develop it themselves

Technological push refers to industry-specific processes, practices, and operations that motivate firms to upgrade their current technological capacity (Horbach et al., 2012; Rennings, 2000). Market pull and technological push are interlinked and complementary, and a firm's capacity to capitalize on R&D influences its trajectory (Kemp et al., 1992). Firms' technological research and development capabilities tend to show a positive relationship with eco-innovation because of their role in facilitating the technological adaptations that are necessary to evolve clean technologies (Bitencourt et al., 2020). Firms have started increasing their investment in environmental innovations to better compete in the marketplace, however, to rationalize these investments in R&D and implementation, an appropriate prospect of future market relevance must be present (Azzone & Noci, 1998). Another aspect of technological push is the drive to develop potential benefits of knowledge spillover within industries; as other firms in the market innovate, firms with strong technological competence are able to benefit from the knowledge gained in order to advance their own eco-innovation (Cai & Li, 2018).

Del Río et al. (2016) concluded that firms in low-tech sectors (slow movers/adopters, mature industries, high polluters) have a higher proclivity to eco-innovate when catalysts in the business environment necessitate for change. For example, innovation may be required to stay within regulations, and thus a low-tech firm which may not have otherwise chosen to eco-innovate would be forced to do so.

In the shipping industry, if shipowners want to eco-innovate, they may send an order to shipyards to build or retrofit ships with existing eco-innovations, or invest in R&D. However, the consequent



improvements in emissions and energy efficiency are dependent on the type of technology, as are the costs (both capital investment and operational). Thus, shipbuilders and suppliers make financial decisions during the innovation phase within this industry which are felt throughout the shipping markets (Köhler & Senge, 2012). The impetus from shipowners to innovate tends to come from demonstration projects that highlight the effectiveness of new technologies. The underlying supposition is that the results from demonstration projects, including both successes and failures, should not reduce shipowner expectations but instead emphasize that learning accumulates during each period, and thus expectations can increase over time (Karslen et al., 2019). It can therefore be argued that technology push is not as significant a driver in adoption as the market push from consumers and the cooperation with authorities in demonstration projects. However, in a recent shipping magazine (Osler & Farley, 2019) it was stressed that WPTs, such as kite sails and Flettner rotors, require a cost-benefit analysis predicated on a vigorous dataset before shipowners are sufficiently well-informed to make large commercial decisions on future adoption.

Proposition 4: Upfront retrofitting costs will be the biggest barrier to adoption

For shipowners, the installation of Flettner rotors is still not profitable enough to justify their cost on all routes with current fuel prices and regulatory policies. Although the environmental gain is substantial, the economic reality is that this technology does not currently make financial sense in situations where wind conditions are less than optimal (Talluri et al., 2016). Additional barriers within the shipping industry include imperfect information on technological performance and the split incentive structure which exists in the short-term time charter market. This last point, for example, implies that shipowners' investments in novel technologies are not rewarded with charter premiums, unless WPTs can become part of the negotiation between the shipowner and the (longterm) user. Other barriers relate to capital access (or a lack thereof), incompatible infrastructure, technical risks, and geopolitical risks such as oil price volatility or lower fleet utilization (Karslen et al., 2019; Rehmatulla, Parker, et al., 2017; Rojon & Dieperink, 2014). These limited charter premiums, as well as low fuel costs and low expectations for benefits, have led shipowners to view rotor technology as uneconomical, leading to low adoption rates and limited learning and to continuing high capital costs. The imperfect information and split incentives barriers prevent knowledge diffusion by stalling positive feedback loops between experiments, learning, and expectations (Karslen et al., 2019). Due to the inadequate mobilization of functional resources, the financial support provided for technological development has primarily been funded through public innovation initiatives, while actors seeking to receive financial support have struggled to obtain it (Rojon & Dieperink, 2014).

Many players in the shipping industry struggle for commercial vitality and oftentimes lack the requisite financial means to invest in, adopt, and subsequently install novel technologies. Within the current market climate, liquidity is more vital to firm survival than profitability, and thus firms tend to avoid investments in which the capital costs would not be recouped for years; thus, any long-term competitive advantage from WPT is overshadowed by the interim capital restrictions (Rojon & Dieperink, 2014).

Given this economic context, one key approach to improving overall adoption rates of WPT may be to decrease capital costs. Centralized approaches such as one-stop retrofit harbors are one possible approach to achieving this. In a 2017 interview, Per Winther Christensen, Deputy Technical Director of the Danish Shipowners' Association, stated that

"I see increasing interest in the Flettner rotor, and Maersk Tankers will install two rotors on a vessel this year. The expected savings on the rotor systems vary and are particularly



dependent on the vessels' trade patterns and which wind systems they meet on the voyage. Savings could be up to 10%, which is significant. Few Flettner rotors are installed and in operation today, but high growth seems likely; if 500 are installed in 2050, that would represent 100-fold growth over 30 years." (Kinthaert, 2017).

While retrofitting costs may be expensive, it may make more sense for new ships to implement WPT, such as Flettner rotors, in the coming decades.

4.2 Economics of shipping firms' operations

In this section, we review the microeconomic decisions of an individual shipping company. The shipping market experiences periods of boom, recession, and depression based on changes in supply and demand. Good microeconomic decision-making means maximizing returns on investment and growth in boom periods, while keeping control of the business when surplus capital is moving out of the industry during a recession. The challenge is to create sufficient profit for the company when times are good to be able to avoid undesirable situations when times are bad. Good decisions give a company a positive cash flow and enable it to implement new technologies to increase profit in future operations. In this section, we review a classic decision-making tool – the return-on-investment model – supporting shipping firms to decide on whether to invest in a set of new wind-assisted propulsion devices.

Return on investment (ROI) provides a snapshot of profitability and can provide a simple quantitative measure of whether or not WPT-equipped ships economically outperform traditional ships. In business, ROI is commonly used to measure rates of return per period for capital invested in an economic entity to decide whether or not to conduct a specific investment. It can also be used as an indicator to compare different investment opportunities. In our case, each shipowner faces the choice between three different technology providers, providing fixed wing rigs, Flettner rotors, or Ventifoils. The investment with the largest return is typically given priority. The ROI can be estimated using equation (1) below (Friedlob & Plewa Jr, 1996). In our study, we rewrite this as equation (2); this will be further developed based on the case study in Section VI.

$$ROI = \frac{Profit}{Investment}$$
(1)
$$ROI = \frac{R_{Invest.} - C_{Invest.}}{C}$$
(2)

Where $R_{Invest.}$ denotes revenue/gains of the investment or reduction of operational costs. $C_{Invest.}$ denotes the cost of the investment.



CInvest.

5. Empirical insights into WPT adoption drivers in the shipping industry

This section presents the case study conducted in Phase 3 of the research, following the theoretical work undertaken in Phases 1 and 2. This case study, which follows the methodological framework set out in Section II above, aims to understand in context how the shipping industry adopts WPTs by presenting the results of a business case workshop and survey of WPT industry and academic experts. The purpose of this section is therefore to understand how WPTs are perceived by shipping stakeholders and to contextualize the shipowners' decision to install them. Phase 3 is a preliminary step in the development of a conceptual model of the adoption of WPT, which is presented in Section VI of this report.

5.1 Business case workshop

The WASP business case workshop was held in May 2020, with the participation of WP4 partners and shipping companies installing WPTs. The summary below sets out the connections between this workshop and the theoretical model development.

The business opportunity

The first part of the workshop addressed the expectations of the shipping companies participating in the WASP Project. A clear concern raised by Shipping Company 1 the learning effects of the installation of WPT in the overall operation of the ship's costs in the longer term. Fuel-saving aspects are therefore critical, especially the possibility to earn back the investment in the long-term (i.e. ten years) (as raised by Shipping Company 2).

Cost-saving concerns go hand in hand with recent regulatory changes in the North Sea Region that require the use of low-sulfur fuels. WPTs are perceived as an emerging alternative that will help achieve both regulatory compliance and fuel cost-savings in the long term (Shipping company 1):

Participation in pilot projects on cleaner technology development is considered a strategic bet by participating companies. Pilot project participation allows a company to build on the learning they have achieved through participation in previous similar projects (Shipping Company 1). Areas where continued learning may be necessary include, for example, the operation period of the technology during voyages (Shipping Company 1). Alongside the technical learning connected to the project, shipping companies value the effect of the cooperation that accompanies participation: "We can help other shipowners through our installation of wind power technology" (Shipping Company 1). Firms also value learning regarding the operational environment once the WPT is installed, in particular, better understanding of the consequences of WPT installation on crew training and cargo handling (Shipping Company 1).

Potential business benefits also exist for the shipping firms in connection with the added brand value of installing WPT on board. Visibility and customer awareness of the investment is important: "The visibility is important, our customers know who you are and know that you are a sustainable carrier, complying with environmental regulations" (Shipping company 1).

Firms expect the installation of WPTs, even to generate financial benefits, even if this installation is only done in the context of pilot projects, as it serves to: "attract more customers and thus impact revenues" (Shipping company 2).



Customers are increasingly demanding that the environmental impacts from shipping operations be addressed (Shipping Company 2). As the participants in the workshop outlined:

"Transport companies ask for CO₂ calculations. They also ask to integrate hybrid propulsion systems. We have many customers asking questions about wind propulsion technologies." (Shipping Company 1)

Challenges, risks, and costs associated with WPT adoption

Participants expressed some concerns about the effects of the COVID-19 pandemic on green transition in the shipping industry. It is not surprising that this context emerged during the workshop, as it took place in May 2020, during the early stages of the pandemic, when lookdowns and uncertainty were on the rise. These uncertainties were connected to fluctuating fuel prices, which made it more difficult to predict costs. This and other conditions resulting from the pandemic led to changes in how much companies were able to prioritize investment in green technologies.

"The current situation is not really stable, due to the COVID-19 pandemic, and our company will postpone investments for at least the next year." (Shipping company 1)

"The current context has a negative effect – green technology is not a priority now. However, this situation may change soon. We perceive that the price [of fuels] is very volatile right now." (Shipping company 2)

In addition to the contextual factors highlighted above, the shipping companies involved in this workshop identified the broader challenges and risks of WPT uptake. The most commonly mentioned was often referred to as "internal" and connected to bunkering costs (Shipping company 1). As discussed in relation to the effects of the pandemic, fuel costs are included in the long-term financial planning of operations, and ongoing uncertainty about these costs implies extra risks. As the participants clarified:

"The challenge right now is the market and low prices paid to freight companies, which extends the length of time taken to recover initial investment" (Shipping company 1).

"The bunker cost is really important and volatile. There is an upward trend in prices over the long term, but in the last few years, prices have gone down. These changes change the business case, especially when the investment in WPT has a payback period of 3 to 5 years" (Shipping company 2).

The installation of WPT incurs additional costs, including costs relating to integrating the routine learning of the new technology into existing procedures: "Operational costs are a major factor when choosing a new technology to install on board ships. Certainly, challenges remain to integrate new technologies with mature ones" (Shipping company 1). In comparison to conventional technologies, in particular fossil-fuel-based solutions, maintenance costs were not considered a key concern (Shipping company 1 and Shipping company 2). However, the associated risks of the installation and potential failures were raised, and one of the shipping company representatives discussed the need for these to be addressed in contracts with WPT providers: "When preparing the contract with the suppler, it is important to leverage the risk and liabilities with respect to the performance of the new technologies" (Shipping company 2).



Another factor to include in consideration of costs is resale value: "WPT can be put on deck, so it will be easy to remove. The resale value of a specific piece of equipment depends on the market acceptance" (Shipping company 1).

5.2 Survey results

The survey results are summarized below according to the questionnaire sections (see Appendix 3).

5.2.1 Environmental performance

Survey questions one, two, and three were used to ascertain the extent to which shipowners embrace environmental performance measures. Question one asked the participants if ecoinnovations such as WPT are necessary to achieve high levels of environmental performance. A resounding 18 of the 19 individuals responded with complete agreement, while one neither agreed nor disagreed. This result is unsurprising since the essence of eco-innovations is their capacity to impact the environment in a positive way. However, it is interesting to note that this question was constructed in such a way as to imply the environment as a stakeholder as opposed to simply an externality. This concept is embedded within the concept of sustainable development and facilities the transition from a utilitarian framework to an ecologically responsible one. Especially for an industry such as shipping, where GHG emissions are a major concern, a major transition is needed if vital climate goals such as those expressed in the Paris Climate Agreement are to be achieved. Eco-innovations such as WPT will go a long way towards reducing fuel consumption and ultimately the amount of pollution caused by this industry.

Question two asked whether WPTs should be adopted by shipowners to strengthen a firm's environmental management strategy. 18 of the 19 respondents either agreed completely or somewhat, while only one neither agreed nor disagreed. Placing this in the context of the analysis of question one, it is increasingly becoming imperative for firms, particularly heavily polluting firms, to develop and implement environmental management strategies. This is not only necessary only from an environmental standpoint but also from an economic perspective; the more proactive firms are, the more they can decrease the amount of catch-up they will eventually have to undertake when more stringent regulations are implemented. WPTs can thus provide an economic benefit but also signal to regulators and industry assessors that shipping firms are taking sustainable development seriously.

Question three addressed the potential of WPTs compared to other available solutions. Roughly 79% of the 19 respondents either completely agreed or somewhat agreed that WPTs have the greatest potential of the existing alternatives to provide immediate environmental relief for the shipping industry. There are several options to shipowners that all provide various levels of environmental relief, however, the allure of WPT is the predicted amount of fuel savings, which acts as a double incentive. Thus, WPT seems more attractive than end-of-pipe technologies that only reduce the amount of pollution, and while a combination of different eco-innovations would be ideal, WPTs offer the most immediate environmental relief considering the financial limitations of shipowners. Furthermore, eco-innovative behavior has been shown to facilitate a positive, albeit indirect, influence on the economic performance of firms. This is because while firms need to adapt to environmental changes and regulations, the spillover from investments such as WPT has the potential to improve economic performance. The associated collaboration within cluster networks also leads to more eco-innovations because firms that have a larger cache of environmental knowledge are better positioned to utilize that knowledge in the future through the development of new products and services.



5.2.2 Regulatory push

Questions four and five centered on regulatory inquiries to determine their potential influence on eco-innovation. Question four asked if governments should provide subsidy schemes to shipowners to increase the proliferation and adoption of WPT in the industry. Of the 19 respondents, 16 were either in complete agreement or somewhat agreed, while two neither agreed nor disagreed and one participant somewhat disagreed. Governments around the world often provide subsidies for both mature and emerging industries. More stringent environmental regulation will lead to greater environmental costs in shipping industry operations because shipowners and operators will need to implement environmental management systems as well as invest in emissions-reducing technologies. Market-based instruments instituted by governing bodies may be useful in providing fiscal incentives to achieve the desired pollution reductions while also increasing the proliferation of eco-innovations to improve adherence to the new standards. If the ambitious emissions goals of the Paris Agreement are to be reached, the decarbonization of the shipping industry must accelerate in the coming decades.

Subsidies could benefit the entire industry, but also specifically entice early adopters and disincentivize fence-sitting. Addressing a related point, the fifth question asked whether there would be an increase in adoption following stricter environmental regulations within the shipping industry. Roughly 90% of the respondents completely agreed or somewhat agreed that there would be, while the other 10% neither agreed nor disagreed. Harsher and more stringent environmental regulations would force the hand of slow-moving shipowners into adopting emissions-reducing technologies like WPT. One option would be universal eco-rating schemes, which can be utilized as a benchmarking instrument for buyers, users, and regulators to make better-informed decisions based on the performance of a given product or service. However, it has been shown that firms can use eco-rating schemes in complicated or obfuscating ways to deflect regulation and/or provide irrelevant information to the marketplace (Poulsen et al., 2018).

A potential countermeasure would be a universal rating scheme that is legislated and overseen by a consortium of national governments. Such a consortium could include the EU and IMO working collectively in the EU as a test region, with the scheme extrapolated and implemented around the world. This scheme should allow for third-party verified data to benchmark the environmental performance of ships to provide reliability. A novel implementation of this could take the form of a decentralized ledger or blockchain to ensure transparency. The potential for regulatory avoidance further emphasizes the necessity for firm regulations. If regulations are definite, then the first-mover advantage is viable, but if regulations are unclear or lax, then a first mover may be at a disadvantage due to premature capital expenditure. This occurred in 2013, when the head of regulatory affairs at AP Moller-Maersk, Niels Bjørn Mortensen, stated that "early movers are getting penalized and all those laughing on the fence are getting rewarded" (Eason, 2013). In this instance, legislation was implemented with a five-year rollout window before being enforced; however, due to lobbying, that deadline was pushed back, thus rendering Maersk's retrofit unnecessarily premature and causing a disadvantage. The spillover from stricter environmental regulations can thus have a positive impact on reducing emissions in the shipping industry, while also proliferating the adoption of eco-innovations such as WPTs.

5.2.3 Market pull

Questions six, seven, and eight were designed to assess market drivers from the perspective of shipowners. Question six asked whether, if one shipowner were to retrofit their ship with WPT, others in the industry would follow; 14 respondents either completely or somewhat agreed that they would, while four respondents saw no impact, and one somewhat disagreed. While this result aligns with the existing literature, it contradicts the series of events outlined above, in which Maersk



invested in NO_x-regulation compliant engines in some vessels, only to have competitors sit on the sidelines while the regulatory organization pushed back the deadline for compliance (Eason, 2013). In this scenario, the rivals did not follow, and as such experienced an advantage due to foregoing capital investments. Although these investments will need to be made at some point if compliance is to be maintained, pushing the investment off gives the advantage to late adopters. They benefit because their ships will not sit idle during retrofitting and they will have more time to secure financing for these retrofits, or even receive subsidies if these were to become available. However, shipping companies may want to push forward with the adoption regardless to differentiate themselves from competitors and take advantage of the public relations benefit of "greening" their operations. In any case, NO_x-regulation compliant technologies do not provide enough of an economic incentive to companies for them to invest in this technology early on the basis of direct economic benefits alone. WPTs, on the other hand, offer the incentive of reduced fuel consumption under certain conditions.

Question seven asked if the upfront costs of retrofitting older ships with WPT outweighed the future environmental and financial cost savings. Eight participants either completely agreed or somewhat agreed that they did, ten respondents were not swayed either way, and one person completely disagreed. Due to many respondents not agreeing nor disagreeing, it is difficult to assess the implications of the responses to this question.

Question eight asked the participants if eco-innovations such as WPT give shipowners a competitive advantage over their rivals; 13 either completely agreed or somewhat agreed that they do, while four did not agree nor disagree, and two somewhat disagreed. The responses to this question suggest that WPT adoption does indeed increase competitiveness. This advantage can take the form of more attractive contract terms with charters due to enhanced fuel efficiency. In any case, this is one of the aspects that the WASP Project aims to investigate. To facilitate generalization, the responses should be triangulated with additional data emerging from the WASP Project in the future,

Furthermore, the public's perception of the shipping industry moving in a "greener" direction bodes well for future business and industry wellbeing. An added positive externality would be the environmental restoration achieved through decreased fuel consumption and increased environmental sustainability.

5.2.4 Technology push

Questions nine and ten were designed to measure technological driving factors. Question nine asked if shipowners' decisions were influenced more by market demands than by technological capabilities; 17 of the 19 respondents completely agreed or somewhat agreed that they were, signifying that while the availability of technology may be important, market conditions and regulatory frameworks have greater influence for decision-makers. Consequently, participation in demonstration projects may be attractive because the exploration costs are shared by numerous parties, as are the potential benefits. In mature industries like the maritime shipping industry, resistance to change is a possibility (Makkonen & Inkinen, 2021), so as a mitigation solution, fewer resources are focused on new and upcoming technologies if development comes at the cost of further market concentration. Ships taken out of service for retrofitting or technological upgrades are rendered idle, causing the firm to lose out on contracts and service time.

Question ten asked if respondents felt WPTs were more suited to short sea routes than long-haul routes. Just under half felt that they are not; only five agreed that they are, while six neither agreed nor disagreed. This somewhat contradicts that prior literature stating that, from the perspective of



adoption, short sea route ships should adopt WPT before longer routes (Rehmatulla, Parker, et al., 2017).

The technology-focused survey questions showed that the capabilities of firms would not significantly impact the adoption rate of WPT. One of the reasons for this is that certain barriers surrounding the implementation of WPT are predicated on a lack of knowledge of the technology itself in practical settings. Demonstrations and trials are therefore vital to the entire network of the maritime industry. While innovation and technological developments play an important role, the awareness that these innovations may present environmental and societal benefits is rapidly increasing, making the economic case stronger than the technological one. The competitiveness between firms in "greening" their operations is contingent upon firms attempting to maintain a solid public reputation with stakeholders and also gaining an advantage over their competitors at both the regional and national levels (Bossle, Dutra de Barcellos, et al., 2016).



6. Proposed theoretical model of WPT adoption in shipping

The development of the agent-based model within this report follows the method discussed in section III. Based on discussions with the programmer overseeing the model coding, a top-down design is used in this project, meaning the entire conceptual model was developed before coding. In the top-down design process, the research question is chosen first, then the agents are identified, and finally, their properties and the rules of their behaviors are designed.

6.1 Choosing the research question

The objective of this report is to provide a theoretical basis for a model of the adoption of windassisted ship propulsion technology by shipowners. There are two keywords in this objective: *windassisted ship propulsion technology* and *shipowners*. Several WPT technologies are available on the market; in this report, we consider fixed wing rigs, Flettner rotors, and Ventifoils. This decision to model multiple WPT suppliers is analogous to Zhang et al.'s (2011) approach, which modelled the characteristics of different alternative vehicle fuels. Our approach to modeling also incorporates multiple types of shipowners operating in different segments; RoPax ferries, bulk carriers, and general cargo vessels are considered for installation of WPTs. We also attempt to integrate the need for shipowners to consider the future market situation (e.g. oil prices) and regulatory conditions (e.g. change in fuel switching costs). Together, these considerations enable us to redefine the research question to *"Can we model the installation of different types of WPT over time by shipowners in different segments with a geographical scope through oscillating market and regulatory conditions?"* To generate results for different market segments, the model may need to run several rounds based on different inputs.

6.2 Choosing mobile agents

The model simulates the decision-making process of shipowners in terms of whether to install and use WPT on their ships. In order to model this complex process, a number of assumptions are made. We assume that each shipowner owns only one ship. Lobbying from technology providers and feedback from other shipowners who use or have used WPT influence the attitude of a shipowner towards WPT. When the attitude reaches a threshold, the shipowner starts to consider whether they need the new technology. When making their decision, the shipowner is driven solely by seeking the highest return on investment. Considering the three types of WPT considered in this report, we include three types of technology providers in this shipowner-technology provider model. Each technology provider promotes one type of WPT. Therefore, in total, four types of agents interact in this model: shipowners and the technology providers for the Flettner rotor, the Ventifoil, and fixed wing rigs.

6.3 Choosing mobile agent properties

Our model includes two general properties which apply to both shipowners and technology providers: (1) location, which records where a single shipowner or technology provider is in the simulation, and (2) heading, which shows the direction of movement of a single shipowner or technology provider. When a shipowner meets a technology provider, the latter attempts to persuade the former to install WPT, and when two shipowners meet, they share their perspectives on WPT. When two technology providers meet, nothing happens. WPTs differ in their technical properties, as summarized by a previous WASP report (Chou et al., 2021). Therefore, we also consider the WPT technical properties as analogous to Zhang et al.'s (2011) characteristics of EVs. Shipowners have two more properties. The first is if cash is not enough to install WPT. Cash



available determines whether a shipowner has enough money to install WPT. If the cash is not sufficient, they may save money during operation. Therefore, we have the second shipowner property, growth rate, which describes the difference in cash available between two consecutive months for a single shipowner if WPT is not in use.

6.4 Choosing environmental agents

In this section, we define the economic environment in which the mobile agents operate. There are two environmental agents included in our model: market situation and regulatory condition. The "fuel price changing rate" used to define the market situation. This is because the main advantage of WPT is cost savings in terms of fuel consumption. Regulatory condition is signified by the change rate of fuel switching costs, in order to simulate the impact on running costs of future regulatory changes (e.g., the possibility that the IMO may expand the area of the Arctic heavy fuel ban)

6.5 Choosing agent behaviors

In this shipowner-technology provider model, we define several behaviors for each agent. These behaviors are necessary for users and programmers to understand how agents interact with each other and the business environment.

Shipowners and technology providers move. The shipowners can turn randomly and move forward. This behavior simulates the social activities of shipowners. Two shipowners or one shipowner and one technology provider moving into one single grid cell in the model signifies that they meet either physically (e.g. participate in an industrial conference) or virtually (e.g. phone calls, online meetings with video link, etc.).

Shipowners interact with shipowners and technology providers. Each shipowner has an initial confidence level regarding WPT. This confidence level will be influenced when a shipowner exchanges information with other shipowners or with technology providers.

Shipowners make decisions on whether to install WPT devices. If the confidence level of a shipowner reaches a threshold, the shipowner consults three technology providers consecutively to choose the most suitable WPT for their ship. One month is considered a reasonable time interval for a shipowner to assess one type of WPT. After-decision making, one year is required to install the WPT equipment¹. The confidence level is expressed on a scale from 0 to 1, with the threshold for installation of WPT set at 0.5. ROI is a commonly used decision-making tool before any type of investment; to calculate ROI, we modify the formula set out earlier in equation (2). Gains from investment, in this case, fall into two categories: total fuel cost savings during the use period of a WPT device (V_{FS}) and the WPT scrap value at the end of life (V_{SV}). Cost of investment also includes two sub-items: the initial amount of investment (C_{IVI}) and the total maintenance cost during the whole use period (C_M). The service life of a WPT device is assumed to be around 30 years². After estimating the present value (PV) of all the future benefits and costs to make all the future cash flows at different times summable, we can calculate the return on investment according to equation (3) below.

 $ROI = \frac{\sum_{t=1}^{T} PV(V_{FS_t}) + PV(V_{SV}) - C_{IVI} - \sum_{t=1}^{T} PV(C_{M_t})}{C_{IVI} + \sum_{t=1}^{T} PV(C_{M_t})}$

(3)

² <u>https://www.stormgeo.com/solutions/shipping/articles/flettner-rotor-sails-for-ship-propulsion/</u>



¹ This period of time was clarified as part of the WASP business case workshop discussion and it is taken as time reference for the theoretical modeling.

Where $PV(V_{FS_t})$, $PV(V_{SV})$ and $PV(C_{M_t})$ are the present values of V_{FS} , V_{SV} and C_M , respectively, in month t. the model runs from month 1 to month 360 over 30 years. T=360. The shipowner will install the WPT devices with the highest ROI.

Shipowners make decisions on whether to use the equipped WPT devices. While WPT reduces ships' energy consumption, it increases shipowners' maintenance costs. Considering that the maintenance cost may decrease if the WPT devices are used less frequently, we split C_M into two sub-items: $C_{M-basic}$ denotes the fixed part of maintenance cost that has to be paid by the shipowners even if the devices are not in use, and $C_{M-marginal}$ denotes the extra cost when the WPT devices are in use. In addition, because fuel prices are not static, shipowners need to estimate V_{FS} at the beginning of each month. They use WPT devices if V_{FS_t} is greater than $C_{M-marginal_t}$; otherwise, they leave the devices idle.

Update cash available. The cash available to a shipowner changes during operation. Therefore, we need to update cash available (CA) at the end of each time interval (t). Regardless of whether WPT devices are installed, shipowners make profits and increase cash available during operations at the speed of ρ_{Growth} . This is the only method for shipowners to save money if their initial cash available is not enough to install WPT devices. After WPT devices are adopted, CA is also influenced by fuel cost savings and maintenance cost. In the year of installing WPT devices or the month of selling the scrap, we also need to subtract C_{IVI_t} or add V_{SV_t} .

$$CA_{t+1} = CA_t \times (1 + \rho_{growth}) + V_{FS_t} - C_{M_t} - C_{IVI_t} + V_{SV_t}$$
(4)

Update confidence level. Shipowners' confidence level will increase if shipowners decide to use WPT devices during the following month or meet technology providers or shipowners who are using WPT; they decrease if shipowners meet other shipowners who have installed WPT devices and do not use them now. They may also undergo small random changes if shipowners meet other shipowners who have never used WPT.



6.6 Designing a time step



Figure 5 Time steps of the interval

6.7 Defining the parameters of the model

We define two parameters in the shipowner-technology provider model: (1) the initial cash available and (2) the initial confidence level. These two parameters enable us to see how different values of the initial setting influence the diffusion of WPT.

6.8 Defining measures

Considering the research question, we define the following measures for each studied ship type:

- The percentage of fixed wing rigs, Flettner rotors, and Ventifoils installed over time under the different scenarios.
- The percentage of fixed wing rigs, Flettner rotors, and Ventifoil in use over time under the different scenarios.
- The fuel price over time under different scenarios.



Stricter regulations in the future may require the use of cleaner types of fuel. Cleaner fuel typically means higher cost, which is reflected by the "change rate of fuel switching costs" in the model. According to the corresponding time interval of a specific fuel price, we are able to check the fuel type from the model. Based on these measures, we can explore the influence of fuel price and regulations on the diffusion and use of WPT.



7. Conclusion

This report addressed two interrelated needs of the WASP Project:

- Designing a theoretical framework to explain wind propulsion technology (WPT) adoption
- Influence of WPT drivers in the context of short sea shipping in the North Sea Region.

This report is part of the WP4 – Business and Policy, activity 5a of the WASP Project, and it will be the basis for the development of an ABM to simulate the adoption of WPT in the context of short sea shipping in the North Sea Region under a range of policy and socio-economic scenarios. The results of this simulation will be delivered in a follow-up report (development of WPT adoption scenarios, simulation, and validation).

The follow-up report will constitute the final deliverable of activity 5a and will take place between May 2021 and June 2022. As a result, the theoretical model presented in Section VI as a result of this report is still subject to further development and can be assessed as the key input to the follow-up report.

Beyond the concrete application of the results to the next stage of the WASP Project, the findings presented here have external validity on their own. In the field of sustainability and socio-technological studies, there are growing calls to introduce simulation methodologies, including ABM, to understand the adoption of eco-innovation in different sectors.

This report extends previous research on the adoption of eco-innovations in the shipping industry by focusing on the microeconomic decision-making aspects of installing WPT. While previous research has already used ABM in the context of the diffusion of eco-innovations, including WPT, in transportation, we have identified certain gaps and opportunities to extend the existing knowledge.

First, the results and methodology presented here fill a gap in the literature on the ABM methodology, primarily relating to the development of the theory underpinning the models. In this report, we present an example of the abductive approach to theory development in ABM using multiple details and a combination of qualitative and quantitative methodologies.

Second, in the emerging literature on eco-innovation, growing attention is being paid to patterns of sustainability transition in specific industrial sectors. Previous research has analyzed the effects of regulations in air pollution abatement technology, green retrofitting of ships, ballast water treatment technologies, and more recently, adoption of cleaner fuels in shipping. This report contributes to knowledge about the driving factors behind the adoption of wind propulsion in shipping, and in particular to improved understanding of the participants involved in the pilot installations co-financed by the WASP Project (North Sea Region).



Bibliography

- Azzone, G., & Noci, G. (1998). Seeing ecology and "green" innovations as a source of change. *Journal of Organizational Change Management*, *11*(2), 94–111. https://doi.org/10.1108/09534819810212106
- Bitencourt, C. C., de Oliveira Santini, F., Zanandrea, G., Froehlich, C., & Ladeira, W. J. (2020). Empirical generalizations in eco-innovation: A meta-analytic approach. *Journal of Cleaner Production*, 245, 118721. https://doi.org/10.1016/j.jclepro.2019.118721
- Boons, F., & McMeekin, A. (2019). An introduction: mapping the field(s) of sustainable innovation. In F. Boons & A. McMeekin (Eds.), *Handbook of Sustainable Innovation* (pp. 1–26). Edward Elgar.
- Bossle, M. B., De Barcellos, M. D., & Vieira, L. M. (2016). Why food companies go green? The determinant factors to adopt eco-innovations. *British Food Journal*, *118*(6), 1317–1333. https://doi.org/10.1108/BFJ-10-2015-0388
- Bossle, M. B., Dutra de Barcellos, M., Vieira, L. M., & Sauvée, L. (2016). The drivers for adoption of eco-innovation. *Journal of Cleaner Production*, *113*, 861–872. https://doi.org/10.1016/j.jclepro.2015.11.033
- Cai, W., & Li, G. (2018). The drivers of eco-innovation and its impact on performance: Evidence from China. Journal of Cleaner Production, 176, 110–118. https://doi.org/10.1016/j.jclepro.2017.12.109
- Chen, J., Cheng, J., & Dai, S. (2017). Regional eco-innovation in China: An analysis of eco-innovation levels and influencing factors. *Journal of Cleaner Production*, *153*(March 2016), 1–14. https://doi.org/10.1016/j.jclepro.2017.03.141
- Cheng, C. C., & Shiu, E. C. (2012). Validation of a proposed instrument for measuring ecoinnovation: An implementation perspective. *Technovation*, *32*(6), 329–344. https://doi.org/10.1016/j.technovation.2012.02.001
- Chou, T., Kosmas, V., Acciaro, M., & Renken, K. (2021). A Comeback of Wind Power in Shipping: An Economic and Operational Review on the Wind-Assisted Ship Propulsion Technology. *Sustainability*, *13*(4), 1880. https://doi.org/10.3390/su13041880
- Christensen, T. B. (2011). Modularised eco-innovation in the auto industry. *Journal of Cleaner Production*, *19*(2), 212–220. https://doi.org/10.1016/j.jclepro.2010.09.015
- Cruz-Cázares, C., Bayona-Sáez, C., & García-Marco, T. (2013). You can't manage right what you can't measure well: Technological innovation efficiency. *Research Policy*, *42*(6–7), 1239–1250. https://doi.org/10.1016/j.respol.2013.03.012
- Cuerva, M. C., Triguero-Cano, Á., & Córcoles, D. (2014). Drivers of green and non-green innovation: empirical evidence in Low-Tech SMEs. *Journal of Cleaner Production*, 68, 104–113. https://doi.org/10.1016/j.jclepro.2013.10.049
- Debenham, J., & Wilkinson, I. (2006). Exploitation versus Exploration in Market Competition. *Industry & Innovation*, *13*(3), 263–289. https://doi.org/10.1080/13662710600858761
- del Río, P., Peñasco, C., & Romero-Jordán, D. (2016). What drives eco-innovators? A critical review of the empirical literature based on econometric methods. *Journal of Cleaner Production*, *112*, 2158–2170. https://doi.org/10.1016/j.jclepro.2015.09.009
- Díaz-García, C., González-Moreno, Á., & Sáez-Martínez, F. J. (2015). Eco-innovation: Insights from a literature review. *Innovation: Management, Policy and Practice*, *17*(1), 6–23. https://doi.org/10.1080/14479338.2015.1011060
- Doran, J., & Ryan, G. (2012). Regulation and firm perception, eco-innovation and firm performance.EuropeanJournalofInnovationManagement,15(4),421–441.https://doi.org/10.1108/14601061211272367



- Dubois, A., & Gadde, L.-E. (2002). Systematic combining: an abductive approach to case research. *Markets as Networks*, 55(7), 553–560. https://doi.org/http://dx.doi.org/10.1016/S0148-2963(00)00195-8
- Eason, C. (n.d.). *Positively prepared*. Lloyd's List. Retrieved March 8, 2020, from https://lloydslist.maritimeintelligence.informa.com/LL038803/Positively-prepared
- Eason, C. (2013). *Fortune favours the Fence-sitter*. Lloyd's List Maritime Intelligence. https://lloydslist.maritimeintelligence.informa.com/LL040867/Nor-Shipping-2013-Fortune-favours-the-fence-sitter
- Easterby-Smith, M., Thorpe, R., Jackson, P. R., & Jaspersen, L. J. (2018). *Management & Business Research* (6th ed.). SAGE.
- Eiadat, Y., Kelly, A., Roche, F., & Eyadat, H. (2008). Green and competitive? An empirical test of the mediating role of environmental innovation strategy. *Journal of World Business*, *43*(2), 131–145. https://doi.org/10.1016/j.jwb.2007.11.012
- Friedlob, G. T., & Plewa Jr, F. J. (1996). *Understanding return on investment*. John Wiley and Sons.
- Garcia, R. (2005). Uses of agent-based modeling in innovation, new product development research. Journal of Product Innovation Management, 22(5), 380-398..pdf. 617, 380–398.
- Garcia, R. (2007). Co-opetition for the Diffusion of Resistant Innovations: A Case Study in the Global Wine Industry using an Agent-based Model. *Proceedings of the 2007 International Conference of the System Dynamics Society*.
- Gilbert, N., Ahrweiler, P., & Pyka, A. (2007). Learning in innovation networks: Some simulation experiments. *Physica A: Statistical Mechanics and Its Applications*, *378*(1), 100–109. https://doi.org/10.1016/j.physa.2006.11.050
- Hasler, K., Olfs, H.-W., Omta, O., & Bröring, S. (2016). Drivers for the Adoption of Eco-Innovations in the German Fertilizer Supply Chain. In *Sustainability* (Vol. 8, Issue 8). https://doi.org/10.3390/su8080682
- Hermann, R. R. (2017). Drivers for environmental technologies selection in the shipping industry: a case study of the North European Sulphur Emission Control Area. International Journal of Environmental Technology and Management, 20(3/4), 139–162. https://doi.org/10.1504/IJETM.2017.089642
- Hojnik, J., & Ruzzier, M. (2016). The driving forces of process eco-innovation and its impact on performance: Insights from Slovenia. *Journal of Cleaner Production*, *133*, 812–825. https://doi.org/10.1016/j.jclepro.2016.06.002
- Holmgren, J., Nikopoulou, Z., Ramstedt, L., & Woxenius, J. (2014). Modelling modal choice effects of regulation on low-sulphur marine fuels in Northern Europe. *Transportation Research Part D: Transport and Environment*, 28, 62–73. https://doi.org/10.1016/j.trd.2013.12.009
- Horbach, J. (2019). Determinants of eco-innovations at the firm level. In F Boons & A. McMeekin (Eds.), *Handbook of Sustainable Innovation* (pp. 60–77). Edward Elgar.
- Horbach, J., Rammer, C., & Rennings, K. (2012). Determinants of eco-innovations by type of environmental impact — The role of regulatory push/pull, technology push and market pull. *Ecological Economics*, 78, 112–122. https://doi.org/10.1016/j.ecolecon.2012.04.005
- Karslen, R., Papachristos, G., & Rehmatulla, N. (2019). An agent-based model of climate-energy policies to promote wind propulsion technology in shipping. *Environmental Innovation and Societal Transitions*, *31*, 33–53. https://doi.org/10.1016/j.eist.2019.01.006
- Kemp, R., Olsthoorn, X., Oosterhuis, F., & Verbruggen, H. (1992). Supply and demand factors of Cleaner technologies: Some empirical evidence. *Environmental & Resource Economics*, 2(6), 615–634.
- Kesidou, E., & Demirel, P. (2012). On the drivers of eco-innovations: Empirical evidence from the UK. *Research Policy*, *41*(5), 862–870. https://doi.org/http://dx.doi.org/10.1016/j.respol.2012.01.005



- Kinthaert, L. (2017). *What will power shipping in 2050?* Lloyd's List Maritime Intelligence. https://lloydslist.maritimeintelligence.informa.com/LL109256/What-will-power-shipping-in-2050
- Köhler, J. (2019). Advances in modelling sustainable innovation: from technology bias to system theories and behavioural dynamics. In Frank Boons & A. McMeeking (Eds.), *Handbook of Sustainable Innovation* (pp. 310–330). Edward Elgar.
- Köhler, J., & Senge, F. (2012). An Agent-Based Model of Transitions to Sustainability in Deep Sea Shipping. Second International Workshop on Successful Strategies (LCS 2012), 9. http://www.lowcarbonshipping.co.uk/files/ucl_admin/LCS 2012/Kohler.pdf
- Makkonen, T., & Inkinen, T. (2021). Systems of environmental innovation: sectoral and technological perspectives on ballast water treatment systems. *WMU Journal of Maritime Affairs*. https://doi.org/10.1007/s13437-021-00226-2

Marshall, C., & Rossman, G. B. (2006). *Designing qualitative research* (4th ed.). Sage.

- Meyer, S. B., & Lunnay, B. (2013). The Application of Abductive and Retroductive Inference for the Design and Analysis of Theory-Driven Sociological Research. *Sociological Research Online*, *18*(1), 12–23.
- Osler, D., & Farley, W. (2019). Shipping 's green road is littered with regulatory hurdles. *Lloyd's List*, *44*(0), 2–3.
- Poulsen, R. T., Hermann, R. R., & Smink, C. K. (2018). Do eco-rating schemes improve the environmental performance of ships? *Marine Policy*, 87, 94–103. https://doi.org/10.1016/j.marpol.2017.10.006
- Rehmatulla, N., Calleya, J., & Smith, T. (2017). The implementation of technical energy efficiency and CO2 emission reduction measures in shipping. *Ocean Engineering*, *139*(June 2016), 184–197. https://doi.org/10.1016/j.oceaneng.2017.04.029
- Rehmatulla, N., Parker, S., Smith, T., & Stulgis, V. (2017). Wind technologies: Opportunities and barriers to a low carbon shipping industry. *Marine Policy*, 75, 217–226. https://doi.org/10.1016/j.marpol.2015.12.021
- Rennings, K. (2000). Redefining innovation eco-innovation research and the contribution from ecological economics. *Ecological Economics*, *32*(2), 319–332. https://doi.org/10.1016/S0921-8009(99)00112-3
- Rivas-Hermann, R., Köhler, J., & Scheepens, A. E. (2015). Innovation in product and services in the shipping retrofit industry: a case study of ballast water treatment systems. *Journal of Cleaner Production*, *106*, 443–454. https://doi.org/10.1016/j.jclepro.2014.06.062
- Rojon, I., & Dieperink, C. (2014). Blowin' in the wind? Drivers and barriers for the uptake of wind propulsion in international shipping. *Energy Policy*, *67*, 394–402. https://doi.org/10.1016/j.enpol.2013.12.014
- Sáez-Martínez, F. J., Díaz-García, C., & Gonzalez-Moreno, A. (2016). Firm technological trajectory as a driver of eco-innovation in young small and medium-sized enterprises. *Journal of Cleaner Production*, *138*, 28–37. https://doi.org/10.1016/j.jclepro.2016.04.108
- Sarkar, A. N. (2013). Promoting Eco-innovations to Leverage Sustainable Development of Ecoindustry and Green Growth. *European Journal of Sustainable Development*, *2*(1), 171–224. https://doi.org/10.14207/ejsd.2013.v2n1p171
- Shrivastava, P. (1995). Environmental technologies and competitive advantage. *Strategic Management Journal*, *16*(S1), 183–200. https://doi.org/10.1002/smj.4250160923
- Talluri, L., Nalianda, D. K., & Giuliani, E. (2018). Techno economic and environmental assessment of Flettner rotors for marine propulsion. *Ocean Engineering*, *154*(April 2017), 1–15. https://doi.org/10.1016/j.oceaneng.2018.02.020
- Talluri, L., Nalianda, D. K., Kyprianidis, K. G., Nikolaidis, T., & Pilidis, P. (2016). Techno economic and environmental assessment of wind assisted marine propulsion systems. *Ocean Engineering*,



121, 301–311. https://doi.org/10.1016/j.oceaneng.2016.05.047

- Tseng, M. L., Wang, R., Chiu, A. S. F., Geng, Y., & Lin, Y. H. (2013). Improving performance of green innovation practices under uncertainty. *Journal of Cleaner Production*, 40, 71–82. https://doi.org/10.1016/j.jclepro.2011.10.009
- Vierth, I., Karlsson, R., & Mellin, A. (2015). Effects of more stringent sulphur requirements for sea transports. *Transportation Research Procedia*, 8, 125–135. https://doi.org/10.1016/j.trpro.2015.06.048

Wilensky, U., & Rand, W. (2015). An Introduction to Agent-Based Modeling. The MIT Press.

Yin, R. K. (2018). Case study research and applications: Design and methods (6th ed.). SAGE.

- Zabalza Bribián, I., Valero Capilla, A., & Aranda Usón, A. (2011). Life cycle assessment of building materials: Comparative analysis of energy and environmental impacts and evaluation of the eco-efficiency improvement potential. *Building and Environment*, *46*(5), 1133–1140. https://doi.org/10.1016/j.buildenv.2010.12.002
- Zhang, T., Gensler, S., & Garcia, R. (2011). A study of the diffusion of alternative fuel vehicles: An agent-based modeling approach. *Journal of Product Innovation Management*, *28*(2), 152–168. https://doi.org/10.1111/j.1540-5885.2011.00789.x



Appendix 1 – Summary of reference ABM on transportation ecoinnovation. Source: Zhang et al. (2011)

Process	Agent	Agent properties	Environment	Other parameters	Data/equation referred to	Useful in WPT model
1. Manufacturer calculates profit	Manufacturer	Profit function of manufacturers, including an engineering design optimization model that will be used to validate the ABM later. Profit is calculated following the equation based on game theory (1), which includes profit as a function of the quantity (q) of each vehicle sold per manufacturer, multiplied by the difference between the price of the vehicle and the manufacturing costs of the vehicle. The result of this product is subtracted from the investment costs to set up a product line. This equation also includes a value of government penalty (in case standards are not met).		The price per vehicle (p) is set as a function of the manufacturing cost (z) and the value of manufacturing parameters (m). As a result, each model includes specific parameters connected to the vehicle design and small changes to its design, including specific pieces, technology, etc.	Michalek et al.'s (2014) equation for engineering performance, consumer demand and manufacturing costs' "simulated annealing" to seek equilibrium among market agents. Vehicle cost data comes from AVCEM.	Validation equation
2. Equilibrium among manufacturers	Manufacturer	Market (Nash) equilibrium among manufacturers/ competitors. Simulated annealing. This algorithm implies that each individual manufacturer keeps optimizing its profit, yet competitor manufacturer decisions remain constant.				
3. Manufacturer designs vehicle agent	Vehicle	Design parameters based on AVCEM (advanced vehicle and energy-use model). This is an analysis package of EV and traditional vehicles. Four parameters on AVCEM: vehicle design, fuel type, engine power, aluminum content). AVCEM not directly used in the ABM, but the costs			AVCEM (Deluche, 2005)	



		associated with each vehicle design are hard-			
		coded into the ABM.			
4. Consumer chooses an EV	consumer	Purchase decision based on customer preference. The equation models the consumer preference for a particular vehicle from product set (C). The equation implies the exp of the utility of consumer for each vehicle, which is particular for each vehicle design, including fuel use, type of vehicle, charge period (partworth)		Based on Garcia (2007). Choice-based conjoint data	
5. Word of mouth influencing the consumers' network	Consumer	Parameters are estimated using hierarchical Bayes (HB). Continuous heterogeneity, each individual parameter is estimated. MCMC parameters estimated. 10 000 iterations are used to derive individual partworths. Each one of the 10 000 individual agents is initiated according to the partworths emerging from the survey results.	Word of mouth (WOM), value is included in the mode through conjoint data analysis and a large-scale survey targeting 20 000 automobile experts (subscribers to a specialized magazine), which received 9504 answers. In the survey, a question was included about frequency in which experts talk to their network about EVs. Partworth is modelled as a function of WOM, knowledge about fuel, knowledge about maintenance, and knowledge about price. WOM is modelled to determine whether this has indeed an effect on consumer preferences, and whether, as a result, government policies and manufacturer decisions should consider WOM into their decision making. One simplified assumption was made, considering each individual agent to be affected in the same way by the WOM effect.	Word of mouth knowledge (Lenk et al 1996)	
6. Government policies	Government	Government creates policies that influence the consumer or the producers (either to increase consumption of EV or to produce more EV). One policy package is investigated (CAFE: corporate average fuel economy). CAFE sets rules about maximum fuel use for new vehicles, therein targeting mainly manufacturers. On the other hand, CAFE also targets consumers because the incentive to reduce costs associated with fuel consumption. In the ABM, the government			



40

agent is therefore modeled as direct costs to the manufacturing as part of equation (6), "total cost of manufacturing for manufacturing agent". The equation includes a difference between the CAFE standard and the fuel economy of the vehicle. in this case, the government is considered an external agent and is not involved in optimization functions		
functions.		



Appendix 2- Summary of reference ABM on WPT diffusion. Source: Karslen et al. (2019)

Process	Agent	Agent properties	Environment	Other parameters
1- Setup	Shipowner/Dry bulk	Each shipowner is assumed to have a homogenous best practice in relation to rotor technologies.	Market risks are considered constant.	Technical risk is used as a variable to determine the adoption of WPT and reflected into the ABM model through a grid implementation model. It addresses the technical risk associated with structural factors, as well as stability and cargo handling
1- Setup		Each shipowner moves 2.5 cells in a random direction as a result of calculated risk.		All ships chartered during the same period
2-Networking between shipowners	Shipowners	Shipowner expectations (EX) in regard to rotor technology. Measured from 0 (no interest) to 1 (complete interest). If an EX threshold is reached, the shipowner starts networking with other interested shipowners.		Knowledge spillover occurs after the shipowner starts to network with other companies
3. Shipowner interaction with technology providers	Shipowner and technology providers	In addition to networking between shipowners, this process models the interaction between technology suppliers and shipowners. The result is increased interest in the WPT (EX). After the threshold is reached (0.5) the shipowner becomes a "supporter" of WPT and increases networking to support WPT adoption		
4. Shipowner fuel costs			Paid by the time charterer	Ship fuel cost (F)
5. Decision to install WPT	Shipowner	If E(NPV) >0, then the shipowner will install WPT		Calculated using expected net present value (E(NPV)).
	Charterer	Incremental revenues		First element in NPV, Incremental revenues to the shipowner (FPT). This function calculates fuel saves resulting from installing the WPT, passed from charterer to the shipowner as a charter premium.



42

	Shipowner	Maintenance cost	Second element in NPV is E(C), which represents additional maintenance cost to the shipowner from rotor technology adoption, and represents costs multiplied by 2, and deducing shipowner expectations.
	Shipowner	Capital cost	In the case of shipowners, NPV also includes E(K) to account for the shipowner's capital of adopting rotor technology.
	Shipowners	Operational profit	E(OP) represents the operational profit and is calculated as the shipowner charter revenues and the operational costs (constant). The equation also includes the deduction of the potential incremental rotor revenues and incremental maintenance costs from the rotor technology installation. This parameter assumes a dry bulk charter regime; it is assumed that OP is positive if the expected incremental operating profit from rotor technology is greater than conventional propulsion OP.
6: Shipowners determine to use rotor technology	Shipowners	Repeating OP condition over several cycles to ensure profitability over time. The rotor technology can be used if it is profitable, or shipowners can decide to take down rotors from the ships, even if the structure stays there if profitability is proven later.	
7: Realized operating profits of shipowners in a period	Shipowners and charterers		Focuses on realized OP for each period. Parameters in equation (6). This equation includes two possibilities. In the first one, the ABM model indicates rotor technology failure for the specific shipowner modelled. In this case, FTP is excluded, which is the incremental shipowners' revenues, and the value "claim" is introduced to specify underperformance as stated by the charterers. In the second possibility, the rotor does not fail; therefore, the realized operational profit will be equal to the FPT, incremental revenues to the shipowner, and the deduced operational costs.



			In equation (6), the value of FPT is also given by an equation that includes monthly cost of fuel (F), charterer expectation of the rotor technology (EX), shipowner expectations (EX) and realized fuel efficiency (RE). RE is adjusted to reflect a percentage of fuel costs to capture the uncertainty of wind propulsion.
8.Shipowners update knowledge through experiments		Shipowners improve their knowledge about rotor technology, barriers, and infrastructure onboard. This is done through experiments reflected through the variable (KN). Similarly, there is networking between shipowners that implies spillover effects among shipowners connected with each other. As a result, other factors are included in the equations including KN incr (increase of knowledge of experimenting shipowner). In each period, the knowledge increases for each shipowner, capturing this value as KN (total spillover), summing up knowledge spillovers from all experimenting shipowners having a link with the focal shipowner.	
9. Shipowners' interactions with demonstration projects	Demonstration projects/ shipowners	Demonstration projects are provided with the property to move around a predefined grid. If these projects move along the grid, they generate expectations among shipowners. "Dradius" defines the degree to which the demonstration project increases knowledge about rotor technology. Other properties include the increase in expectation and knowledge of shipowners fitting within the demonstration project radius (DEX incr). Other properties include the assumptions that demonstration projects are successful once implemented, thus assuming they do not decrease shipowners' interests and expectations	
10. Technical risk update	Shipowners		This focuses on the risk associated with the rotor technology and is given by an inverted s-shaped curved equation 9 in the model. The risk factor is calculated according to the difference to the KN factor (shipowner



			knowledge about rotor technology). KN increases if the shipowners interact in a network together with other actors involved in demonstration projects.
11. Rotor technology capital costs		This process captures an environmental characteristic, namely that of the installed capacity of the rotor technology. This implies that as more ships install rotors –also as demonstration projects – then the capital cost installation of one rotor piece will also decrease proportionally. The model assumes a double capacity to account for a reduction of the capital cost (LR) of a new installation.	
12. Shipowners and charterers update expectations	Shipowners		The shipowner's expectation (EX) is updated after niche experiments following the original expectation and realized operational profit (OP).



Appendix 3- Survey with experts

Online Survey

- a. What is your name? (First and surname)
- b. What is your email address?
- c. Country
- d. Type of organization
- e. Position or title within the organization *(optional)

On a scale of 1-5, to what extent do you agree with the following statements?

Environmental performance

1. Eco-innovations in the shipping industry, such as wind propulsion technologies, are necessary to achieve high levels of environmental mitigation performance.

2. Wind propulsion technologies should be adopted by shipowners to strengthen a firm's environmental management strategy.

3. Of the solutions available, wind propulsion technologies have the potential to provide the most immediate environmental relief for the shipping industry.

Regulatory push

4. The government should provide subsidy schemes to shipowners to increase their adoption of wind propulsion technology.

5. Stricter environmental regulations within the shipping industry would lead to an increase in wind propulsion technology adoption by shipowners.

Market pull

6. If one shipowner retrofits with wind propulsion technology, others within the industry will follow.

7. In regard to retrofitting older ships with wind propulsion technologies, upfront costs outweigh future environmental and financial cost savings.

8. Eco-innovations like wind propulsion technologies improve shipowners' competitive position in the market over rivals.

Technology push

9. Shipowners are more influenced by market demands than the technological capabilities of the firm.

10. Wind propulsion technologies are more suited for short sea routes than long hauls

